SPECIFIC COMMENTS TO AUTHORS

The manuscript by Schinas et al discusses a very important problem in the healthcare sector. With the rising incidence of drug-resistant fungal infections and the declining novel antifungal discovery, the problem and the solutions are highlighted in this article properly. Overall, the manuscript is very well written and presented in a lucid manner. I do not have any specific comments on the manuscript.

Answering Reviewers: thank you for your kind comments.